RESEARCH

Open Access



A behavioral test battery for mouse models of Angelman syndrome: a powerful tool for testing drugs and novel *Ube3a* mutants

Monica Sonzogni^{1,2†}, Ilse Wallaard^{1,2†}, Sara Silva Santos^{1,2†}, Jenina Kingma^{1,2}, Dorine du Mee^{1,2}, Geeske M. van Woerden^{1,2} and Ype Elgersma^{1,2*}

Abstract

Background: Angelman syndrome (AS) is a neurodevelopmental disorder caused by mutations affecting UBE3A function. AS is characterized by intellectual disability, impaired motor coordination, epilepsy, and behavioral abnormalities including autism spectrum disorder features. The development of treatments for AS heavily relies on the ability to test the efficacy of drugs in mouse models that show reliable, and preferably clinically relevant, phenotypes. We previously described a number of behavioral paradigms that assess phenotypes in the domains of motor performance, repetitive behavior, anxiety, and seizure susceptibility. Here, we set out to evaluate the robustness of these phenotypes when tested in a standardized test battery. We then used this behavioral test battery to assess the efficacy of minocycline and levodopa, which were recently tested in clinical trials of AS.

Methods: We combined data of eight independent experiments involving 111 *Ube3a* mice and 120 wild-type littermate control mice. Using a meta-analysis, we determined the statistical power of the subtests and the effect of putative confounding factors, such as the effect of sex and of animal weight on rotarod performance. We further assessed the robustness of these phenotypes by comparing *Ube3a* mutants in different genetic backgrounds and by comparing the behavioral phenotypes of independently derived *Ube3a*-mutant lines. In addition, we investigated if the test battery allowed re-testing the same animals, which would allow a within-subject testing design.

Results: We find that the test battery is robust across different *Ube3a*-mutant lines, but confirm and extend earlier studies that several phenotypes are very sensitive to genetic background. We further found that the audiogenic seizure susceptibility phenotype is fully reversible upon pharmacological treatment and highly suitable for dose-finding studies. In agreement with the clinical trial results, we found that minocycline and levodopa treatment of *Ube3a* mice did not show any sign of improved performance in our test battery.

Conclusions: Our study provides a useful tool for preclinical drug testing to identify treatments for Angelman syndrome. Since the phenotypes are observed in several independently derived *Ube3a* lines, the test battery can also be employed to investigate the effect of specific *Ube3a* mutations on these phenotypes.

Keywords: Angelman syndrome, UBE3A, Mouse model, behavior, drug screening

* Correspondence: y.elgersma@erasmusmc.nl

²ENCORE Expertise Center for Neurodevelopmental Disorders, Erasmus Medical Center, Rotterdam, Netherlands



© The Author(s). 2018 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

 $^{^{\}rm t}{\rm Monica}$ Sonzogni, Ilse Wallaard and Sara Silva Santos contributed equally to this work.

¹Department of Neuroscience, Erasmus Medical Center, Rotterdam, Netherlands

Background

Angelman syndrome (AS) is a neurodevelopmental disorder first described in 1965 by Harry Angelman, with a birth incidence of approximately 1:20,000 [1]. AS is caused by the functional loss of the maternal allele encoding an E3 ubiquitin-protein ligase (UBE3A) [2]. Loss of functional UBE3A results in the core phenotypes of severe intellectual disability, motor coordination deficits, absence of speech, and abnormal EEG, as well as in high comorbidity of sleep abnormalities, epilepsy, and phenotypes related to autism spectrum [3].

Currently, only symptomatic treatments are available for AS, primarily aimed at reducing seizures and improving sleep [4]. The development of targeted treatments for AS heavily relies on the ability to test the efficacy of treatments in mouse models of the disorder. The success of such translational studies depends on three critical factors [5]: (1) high construct validity, (2) high face validity, and (3) robustness of the behavioral phenotypes. First, the construct validity (shared underlying etiology between mouse models and patients) of the AS mouse model is very good, since AS mouse models recapitulate the patient genetics by carrying a mutated Ube3a gene specifically at the maternal allele. However, it should be noted that the majority of the AS patients carry a large deletion (15q11-15q13) which encompasses also other genes besides the UBE3A gene, and which may contribute to a more severe phenotype [6]. Second, with respect to face validity (i.e., similarity of phenotypes between patient and the mouse model), the AS mouse model captures many neurological key features of the disorder really well (e.g., epilepsy, motor deficits, abnormal EEG), as well as some of the behavioral abnormalities (abnormal sleep patterns, increased anxiety, repetitive behavior) [7–12]. Robustness of the behavioral phenotypes is the third important aspect to identify novel treatments, as it allows experiments to be sufficiently powered to detect the effect of the treatment, and meanwhile minimizes a type I error in which a drug is declared effective whereas it is not. Robustness, as well as face validity, also takes into account the sensitivity to genetic background and the extent in which a phenotype is also observed in independently derived mouse models. Notably, almost all behavioral testing described in literature has been performed using the original Ube3a^{tm1Alb} mouse strain generated in the Beaudet lab [7-9]; hence, it is unknown to what extent the reported phenotypes are actually specific to this mouse line.

We previously developed a series of behavioral paradigms in the domains of motor performance, anxiety, repetitive behavior, and seizure susceptibility, for testing the effect of *Ube3a* gene reinstatement in the inducible $Ube3a^{mSTOP/p+}$ (*Ube3a*^{tm1Yelg}) mice [13]. Here, we used these paradigms in a highly standardized way, to assess phenotypes in the independently derived Ube3a^{tm1Alb} and *Ube3a*^{mE113X/p+} (*Ube3a*^{tm2Yelg}) maternal knockout strains. We combined data of eight independent experiments across five experimenters involving 111 Ube3a^{t-} ^{m1Alb} and 120 wild-type littermate control mice. Using a meta-analysis, we determined the statistical power of the different behavioral tests and the effect of putative confounding factors, such as the effect of sex differences. We further assessed the robustness of these phenotypes by comparing Ube3a mutants in different genetic backgrounds. Finally, we employed this behavioral test battery to reassess the efficacy of minocycline and levodopa in the AS mouse model. Minocycline is a matrix metalloproteinase-9 inhibitor (MMP9), a tetracycline derivative which possesses antibiotic as well as neuroprotective activity [14, 15]. Its antibiotic properties against both gram-positive and gram-negative bacteria are related to its ability to bind to the bacterial 30S ribosomal subunit, thereby inhibiting protein synthesis [14].

Levodopa is the precursor of dopamine and was shown to be effective in treating Parkinsonism in two adults with Angelman syndrome [16]. Moreover, it is able to reduce CAMK2 phosphorylation [17], which was shown to be increased in a mouse model for Angelman syndrome [18, 19]. Minocycline and levodopa were previously tested in the AS mouse model and based on the favorable outcome of these preclinical experiments, three clinical trials were performed [20–22]. Unfortunately, none of these drugs showed a significant improvement in AS patients.

Methods

Mouse husbandry and breeding

For this study, we used $Ube3a^{m-/p+}$ mice ($Ube3a^{tm1Alb}$; MGI 2181811) [7] and $Ube3a^{mE113X/p+}$ mutants ($Ube3a^{t-M2Yelg}$; MGI5911277) as previously described [23]. $Ube3a^{tm1Alb}$ mice were maintained (> 40 generations) in the 129S2 background (full name: 129S2/SvPasCrl) by crossing male $Ube3a^{m+/p-}$ mice with female 129S2 wild-type mice. $Ube3a^{tm2Yelg}$ mice were maintained (> 20 generations) in the C57BL/6J (Charles River) background by crossing male $Ube3a^{m+/p-113X}$ mice with female C57BL/6J wild-type mice. For the seizure susceptibility experiments with $Ube3a^{mE113X/p+}$ animals, this line was backcrossed eight times in 129S2 by crossing $Ube3a^{pE113X/m+}$ males with 129S2 wild-type females.

For behavioral experiments, female $Ube3a^{tm1Alb}$ $(Ube3a^{m+/p-})$ mice were bred to yield $Ube3a^{m-/p+}$ mice in two different backgrounds: $Ube3a^{m-/p+}$ (AS) mice and their WT littermates in the F1 hybrid 129S2-C57BL/6J background (WT = 120, AS = 111) and in the 129S2 background (WT = 11, AS = 16). $Ube3a^{mE113X/p+}$ mice and their WT littermates were generated in the same manner in the F1 hybrid 129S2-C57BL/6J background (WT = 10, $Ube3a^{mE113X/p+} = 10$) and in C57BL/6J background (WT = 15, $Ube3a^{mE113X/p+} = 16$).

For the seizure susceptibility test, we used $Ube3a^{m-/p+}$ (WT = 45, AS = 114) and $Ube3a^{mE113X/p+}$ mice (WT = 4, AS = 8) in the 129S2 background.

Mice were housed in individually ventilated cages (IVC; 1145T cages from Techniplast) in a barrier facility. Mice were genotyped when they were 4-7 days old and re-genotyped at the completion of the experiments. All animals were kept at 22 ± 2 °C with a 12-h dark and light cycle and were tested in the light period, provided with mouse chow (801727CRM(P) from Special Dietary Service) and water ad libitum. During behavioral testing, mice were group-housed with two to four animals of the same sex per cage. Fighting between males was observed a few times, and in these rare cases, mice were separated and single housed. This was not a reason for exclusion. All mice were single housed during nest building and for the subsequent forced swim test. All animal experiments were conducted in accordance with the European Commission Council Directive 2010/63/EU (CCD approval AVD101002016791).

Behavioral analysis

The weight of the animals was determined a few days before the start of the behavioral analysis. Prior to each test, mice were acclimatized to the testing room for 30 min.

All behavioral experiments were performed during the light period of the light/dark cycle. Both male and female mice at the age of 8–12 weeks were used for the experiments. Moreover, we tried to obtain a similar ratio of females/males between the WT and AS groups. Only in the experiments described in Fig. 4 (*Ube3a*^{E113X} mice in F1 background) and in the epilepsy experiment using *Ube3a*^{E113X} mice (Fig. 6c), the female/male ratio between the groups was significantly different (p < 0.05; chi-square test).

All behavioral testing and scoring was performed by experimenters who were blind to genotype and treatment. Behavioral tests were always run in the following order and with a minimal number of days between tests: (1) accelerating rotarod test for 5 consecutive days performed at the same hour every day; (2) 2 days of pause; (3) open field test; (4) 1 day of pause; (5) marble burying test; (6) between 5 and 7 days of pause to allow adaptation to being single caged; (7) nest building test for 5 consecutive days, in which the weight of the nest was assessed at the same hour every day; (8) 2 days of pause; and (9) forced swim test.

Accelerating rotarod

Motor function was tested using the accelerating rotarod (4–40 rpm, in 5 min; model 7650, Ugo Basile Biological

Research Apparatus, Varese, Italy). Mice were given two

trials per day with a 45–60-min inter-trial interval for 5 consecutive days (same hour every day). For each day, the average time spent on the rotarod was calculated, or the time until the mouse made three consecutive wrapping/ passive rotations on the rotarod (latency in seconds). These passive rotations were observed rarely (1-2%) in 129S2 or F1 hybrid 129S2-C57BL/6J mice but rather common in (30%) C57BL/6J mice. Maximum duration of a trial was 5 min.

Open field test

To test locomotor activity and anxiety, mice were individually placed in a 110-cm-diameter circular open field and allowed to explore for 10 min. The light intensity was approximately 25–30 lx measured in the center of the arena. The total distance moved by each mouse in the open arena was recorded by an infrared camera (Noldus[®] Wageningen, NL) connected to the EthoVision[®] software (Noldus[®] Wageningen, NL), and the final outcome is indicated as distance moved in meters. For some groups, we also analyzed the time spent in the inner zone (IZ), middle zone (MZ), and outer zone (OZ) (IZ r = 25 cm, MZ r = 40, OZ r = 55 cm).

Marble burying test

Open Makrolon (polycarbonate) cages $(50 \times 26 \times 18 \text{ cm})$ were filled with 4 cm of bedding material (Lignocel[®] Hygenic Animal Bedding, JRS). On top of the bedding material, 20 blue glass marbles were arranged in an equidistant 5×4 grid and the animals were given access to the marbles for 30 min. After the test, the mice were gently removed from the cage. Marbles covered for more than 50% by bedding were scored as buried, and the outcome measured is the number of buried marbles.

Nest building test

To measure nest building, mice were single housed for a period of 5 to 7 days before the start of the experiment. Subsequently, used nesting material was replaced and 11 g (11 ± 1) of compressed extra-thick blot filter paper (Bio-rad©) was added to the cage. The amount of the unused nest material was weighed and noted every day for a consecutive of 5 days, each day at the same hour.

Forced swim test

Mice were placed for 6 min in a cylindrical transparent tank (27 cm high and 18 cm diameter), filled with water (kept at 26 ± 1 °Celsius) 15 cm deep. The mouse was first left in the cylinder for 2 min to habituate. Immobility during the forced swim test was scored manually (stop-watch) by timing the amount of time the mouse was floating in the water (defined by lack of any movement) and was assessed during the last 4 min of the test.

The mouse was considered to be immobile when he ceased to move altogether, making only movements necessary to keep its head above water. The outcome measured is the time in seconds in which the mouse was immobile.

Susceptibility to audiogenic seizures

Because of the different genetic background requirements, an independent cohort of mice was used to test susceptibility to audiogenic seizures. Mice were placed in Makrolon (polycarbonate) cages ($50 \times 26 \times 18$ cm), and audiogenic seizures were induced by vigorously scraping scissors across the metal grating of the cage lid (which creates approximately a 100-dB sound). This noise was generated for 20 s, or less if a tonic-clonic seizure developed before that time. Susceptible mice responded with wild running and leaping followed by a tonic-clonic seizure, which typically lasted 10–20 s.

Within-subject testing

For the experiment described in Fig. 3, *Ube3a^{tm1Alb}* mice in F1 hybrid 129S2-C57BL/6J background were subjected to the behavioral test battery for a second time. Once the first battery was completed, female mice that had been single housed for the nest building test were placed back together with the original cage mates, while male mice remained separated for the entire second set of behavioral tasks. The second test started 4 weeks after the first testing was completed.

Drug administration

Vehicle treatment

All animals used for the meta-analysis were treated with vehicle either by IP injection (max volume 10 ul/g, hypodermic-needle $25G \times 16$ mm (Sterican[®]/B-Braun)), by oral gavage (max 10 ul/g, stainless steel animal feed-ing tubes $20G \times 38$ mm (Instech Laboratories)), or by adding to the drinking water.

Minocycline treatment

The adult-treated group consisted of 8–10-week-old $Ube3a^{m-/p+}$ (n = 11 saline; 11 minocycline) and WT (n = 9 saline; 10 minocycline) littermate control mice in F1 hybrid 129S2-C57BL/6J background. Due to space limitations, only six animals per group were used for nest building. Mice were assigned to two treatment groups in such a way that both groups had a comparable distribution of males and females and mutant and wild-type mice. Mice were subjected to daily minocycline or vehicle IP injections (minocycline hydrochloride, Sigma-Aldrich 45 mg/kg in saline solution), starting 3 weeks prior to commencing behavioral testing, as previously described [20, 24]. Behavioral testing was started 1.5 h post-injection, based on the half-life of minocycline (~ 2 h in plasma), and the

peak brain levels are reached about 2 h after injection [25].

For the postnatal-treated group, cages with $Ube3a^{m-/p}$ + and WT pups in F1 hybrid 129S2-C57BL/6J background were split in two treatment groups in such a way that both groups had a comparable distribution of males and females and mutant and wild-type mice. The treatment group received minocycline via the lactating dam, which received minocycline through the drinking water (0.2 mg minocycline/ml, supplemented with 1 mg/ml aspartame to counteract the bitter taste and shielded for light) [26]. This method of administration was shown to yield detectable concentration of minocycline in the blood of adult mice [27] and in the breast milk of lactating dams [28, 29]. Once the mice were weaned, they were supplied with the same concentration of minocycline in their drinking water. Assuming a water intake of 1.5 ml/10 g body weight/day [30], and assuming an average weight of 25 g/mouse, the average amount of minocycline these mice received is approximately 30 mg/kg/ day. The drinking water was refreshed every other day. Treatment continued until all behavioral experiments were completed. The control group received water with aspartame.

Levodopa/carbidopa treatment

Cages containing *Ube3a^{m-/p+}* and wild-type littermate control mice (8–12 weeks old) in the F1 hybrid 129S2-C57BL/ 6J background were assigned to two treatment groups in such a way that both groups had 15 wild type and 15 mutants and a comparable distribution of males and females. Mice in the treatment group received 15 mg/kg levodopa and 3.75 mg/kg carbidopa dissolved in saline (levodopa, Sigma-Aldrich; carbidopa, Sigma-Aldrich) by IP injection with an injection volume of 10 ul/g. The untreated group received vehicle injection by IP as described by Tan et al. [21]. The mice were injected 1 h prior to carrying out the behavioral tasks, during the entire period while partaking in these tests.

Levetiracetam treatment

 $Ube3a^{m-/p+}$ mice in the 129S2 background were first tested for audiogenic seizure susceptibility at baseline. Minimally 24 h later, the mice were again tested for audiogenic seizure susceptibility, this time precisely 1 h following a single IP injection of levetiracetam (0–0.5–1–2–10–15 mg/kg; Sigma-Aldrich). The injection volume used is 5 ml/kg, and the drug was dissolved in 1% Tween-80 (Sigma-Aldrich) in milliQ water as previously described [31].

Data analysis

Data was analyzed using Excel 2010 (Microsoft) and IBM SPSS software (NY, USA). The open field, marble

burying, and forced swim test data were analyzed using an unpaired T test in the untreated experimental groups and a two-way ANOVA in minocyclineand levodopa-treated animals (in which we assessed a genotype-treatment interaction). Rotarod and nest building were measured with a repeated measures ANOVA in the untreated experimental groups, or with a multivariate repeated measures ANOVA (assessing significance of interaction of time, genotype, and treatment) in the minocycline and levodopa experimental groups. We used a Bonferroni's post hoc test to detect significant differences in male and female groups. For the within-subject experiment, we used a paired T test for open field, marble burying, and forced swim tests, while we used a repeated measures factorial ANOVA when analyzing the rotarod and the nest building test. For the audiogenic seizure analysis, a Fisher's exact test was used. The correlation between body weight and maximal performance on the rotarod test was assessed with a Pearson's correlation test. For the power calculation, we performed a priori analysis using G^{*}Power 3.1 software [32] with $\alpha = 0.05$ and power $(1 - \beta) = 0.95$, 0.90, or 0.80. Data is presented as mean± SEM in all figures. For all tests, statistical significance was denoted by $p \le 0.05$ (*), p < 0.01 (**), and *p* < 0.001 (***).

A chi-square test was performed to test if there were any significant differences in the ratio of females/males between the WT and AS groups.

Results

Robust behavioral phenotypes in $Ube3a^{m-/p+}$ mice in the F1 hybrid 129S2-C57BL/6J background

We recently developed a number of behavioral tests for testing the effect of gene reinstatement in inducible $Ube3a^{mSTOP/p+}$ ($Ube3a^{tm1Yelg}$) mice [13]. These tests can be applied in successive order to assess phenotypes in the domains of motor performance, anxiety, and repetitive behavior. Here, we set out to assess the robustness of these phenotypes in an independently derived mouse model of AS, by using F1 hybrid 129S2-C57BL/6J $Ube3a^{m-/p+}$ ($Ube3a^{tmIAlb}$) mice [7], which is the *Ube3a* mouse mutant used for nearly all behavioral studies. We have frequently used this strain to test the efficacy of novel treatments and combined all data obtained from vehicle-treated $Ube3a^{m-/p+}$ and wild-type littermate controls in the F1 hybrid 129S2-C57BL/6J background to perform a meta-analysis. In total, this constitutes the combined data of eight experiments, carried out by five experimenters and totaling 111 $Ube3a^{m-/p+}$ and 120 wild-type littermate controls (Table 1; Fig. 1).

Individuals with Angelman syndrome show clear motor impairments, and impaired performance on the accelerating rotarod is the most frequently described phenotype in *Ube3a* mice. Indeed, our meta-analysis shows a very robust significant difference between the two genotypes (p < 0.001; Fig. 1a). A power analysis with

Exp. #	Person	WT/MUT (n)	Sex WT f/m MUT f/m (n)	Rotarod (time(s)) WT mean (SD) Mut mean (SD)	Open field (distance(m)) WT mean (SD) Mut mean (SD)	Marble burying (# marbles buried) WT mean (SD) Mut mean (SD)	Nest building (% material used) WT mean (SD) Mut mean (SD)	Forced swim test (% floating) WT mean (SD) Mut mean (SD)
1	A	15/13	8/7 6/7	128 (42) 96 (32)	41 (14) 22 (11)	11 (4) 4 (3)	14 (25) 79 (18)	53 (23) 83 (7)
2	A	15/13	5/10 3/10	142 (43) 80 (32)	49 (10) 32 (12)	8 (4) 2 (3)	36 (23) 79 (14)	44 (24) 81 (7)
3	А	15/13	5/10 3/10	133 (42) 92 (46)	40 (12) 29 (8)	11 (3) 2 (2)	27 (18) 70 (18)	41 (19) 73 (14)
4	В	21/171)	10/11 7/10	159 (60) 102 (36)	31 (12) 19 (11)	14(4) 4(5)	10 (11) 48 (25)	24 (22) 63 (18)
5	С	9/11 ²⁾	5/4 6/5	163 (49) 91 (37)	25 (6) 10 (7)	12 (5) 4(5)	48 (27) 69 (12)	28 (24) 76 (8)
6	D	15/14 ³⁾	4/11 6/8	107 (44) 74 (26)	44 (7) 29 (13)	12 (3) 3 (3)	40 (18) 79 (12)	14 (20) 60(31)
7	E	15/15	8/7 7/8	196 (57) 126(52)	45(10) 35 (7)	11 (5) 6 (3)	63 (20) 74 (14)	47 (20) 67 (14)
8	А	15/15 ⁴⁾	7/8 8/7	162 (49) 95 (35)	49 (9) 33 (13)	10 (3) 2 (3)	N/A	47 (20) 88 (9)

 Table 1 Overview of experiments used for the meta-analysis

All experiments were performed using *Ube3a^{tm1Alb}* mice in F1 hybrid 12952-C57BL/6J background. For all tests shown in this table, we found a significant effect of genotype (p < 0.05), except for the nest building test of experiment 8, which was not performed. The table indicates the individual that performed the test battery, the number of wild-type and mutant mice used for each test, the number of females and males used for each experimental group, and the mean and standard deviation of the outcomes obtained. For the rotarod, we indicated the average performance over the 5 days, while for the nest building we provided the data as measured at day 5. Note that for some of the tests, we used a different number of mice (mice were not properly tracked, or a smaller cohort was used for nest building because of space limitations). The adapted *n* for these experiments is as follows: ¹)nest building 13/12, forced swim test 20/17; ²)nest building 6/7; ³open field 13/14; ⁴open field 10/10, nest building not performed



(See figure on previous page.)

Fig. 1 Behavioral testing of $Ube3a^{im1Alb}$ mice in F1 hybrid 129S2-C57BL/6J background. For each behavioral paradigm, the pooled (raw) data of all experiments is presented on the left panel, whereas the Forrest plots in the middle panel show the normalized data of the individual experiments (in which the data of each experiment is normalized against wild type; represented by a dashed line), as well as the 95% confidence interval. The picture on the right panel shows the behavioral set-up used for our experiments. For the marble burying test and nest building test, the picture shows the onset and finish of a behavioral experiment. **a** Accelerating rotarod in wild-type (WT) and $Ube3a^{m-/p+}$ mice (n = 120, 111). **b** Open field test in WT and $Ube3a^{m-/p+}$ mice (n = 120, 111). **d** Nest building test in WT and $Ube3a^{m-/p+}$ mice (n = 94, 86). **e** Forced swim test in WT and $Ube3a^{m-/p+}$ mice (n = 120, 111). All data represent mean \pm SEM. A repeated measures ANOVA or T test was used for statistical comparison of the non-normalized data. All tests show a significance effect of genotype (***p < 0.001)

 $\alpha = 0.05$; $(1 - \beta) = 0.95$ showed that this task requires 14 animals per genotype (Table 2).

Following 2 days of rest, the same mice were then tested in the open field test. This paradigm is commonly used to assess anxiety in mice. Increased anxiety is commonly observed in individuals with AS [33], as well as individuals with autism spectrum disorder. In this test, we place the mice in an open arena situated in a brightly lit room and record the distance the mice travel during a 10-min time span. The measurements of the distance moved in the open arena indicated that AS mice moved significantly less (WT 40.3 \pm 1.2 m; AS 26.2 \pm 1.2 m; p < 0.001; Fig. 1b). A power analysis ($\alpha = 0.05$; $(1 - \beta) = 0.95$) showed that this task requires a minimum number of 21 mice per genotype, which makes this test a relative weak test (Table 2). Previous studies reported no significant difference observed between genotypes in the time spent in the [8, 9] inner zone of the open field, which is another measure of anxiety. Our meta-analysis revealed a significant difference between genotypes (p < 0.005), but this difference was small (WT 1.1% versus mutant 0.7% time in inner zone), and a significant effect was only observed in four out of the eight individual experiments (data not shown).

Page 7 of 19

After 1 day of rest, the same mice were then analyzed in the marble burying test, a test used to assess repetitive and perseverative behavior as well as anxiety [34, 35]. When exposed to marbles, AS mice show a strongly impaired marble burying behavior compared to WT mice (WT 11.3 ± 0.4; AS 3.6 ± 0.3; p < 0.001; Fig. 1c). A power analysis ($\alpha = 0.05$; (1 – β) = 0.95) showed that seven animals/group are sufficient for this test, indicating a very robust phenotype (Table 2).

After the marble burying task, all mice were single housed for 5–7 days and then analyzed for 5 consecutive days while performing the nest building test. The nest building test assesses the innate behavior of mice to create a nest to maintain body temperature and to find shelter [36]. AS mice showed a clear phenotype compared to their WT control littermates (p < 0.001; Fig. 1d). As indicated in Table 2, the nest building phenotype is quite robust, since it only requires 8 mice ($\alpha = 0.05$; $(1 - \beta) = 0.95$) per group if analyzed over the last day.

Following 2 days of pause, the animals were finally subjected to the forced swim test, in which the mouse is placed in a beaker filled with water, from which the mouse will try to escape by swimming. This test is typically used to test depressive-like behavior in mice [37]. AS mice showed significant more time floating (instead

Tab	le 2	. Achievec	l power fo	or each	behavioral	test of	f the	behav	ioral	test	battery
-----	------	------------	------------	---------	------------	---------	-------	-------	-------	------	---------

	Wild type (mean ± SD)	Ube3a	Test	Achieved effect size	Sample size per group	Sample size per group	Sample size per group	
	(,	(mean ± SD)			$(1-\beta) = 0.95$	$(1-\beta) = 0.90$	$(1-\beta) = 0.80$	
Rotarod Time on machine (s)	149 ± 55	95 ± 40	ANOVA	0.56	14	11	9	
Open field Distance moved (m)	40 ± 13	26 ± 13	T test	1.17	21	17	13	
Marble burying (# marbles buried)	11 ± 4	4 ± 4	T test	2.26	7	6	5	
Nest building (% used nesting material)	68 ± 23	28 ± 19	T test	1.95	8	7	6	
Forced swim test (% floating time)	37 ± 25	73 ± 18	T test	1.73	10	9	7	
Susceptibility to audiogenic seizure (% of animals)	7	98	T test	4.55	3	3	3	

Data provided is based on the experiments using *Ube3a^{tm1A/b}* mice in F1 hybrid 12952-C57BL/6J background. The table provides the obtained effect size, number of mice needed per genotype for each behavioral test (with power equal to 0.95, 0.90, 0.80), and statistical test used. For rotarod calculations, we used the average performance over the 5 days, while for the nest building we used the data of the last test day

of swimming) compared to WT mice (WT 36.8 ± 2.3; AS 72.6 ± 1.7; p < 0.001; Fig. 1e). The power analysis test showed that this task requires a minimum of 10 mice ($\alpha = 0.05$; (1 – β) = 0.95).

Taken together, the data indicates that this test battery yields a series of robust behavioral phenotypes that can be obtained in a relative quick manner using a single cohort of mice.

The dependence of sex on the behavioral phenotypes

Angelman syndrome affects both males and females, with no known differences between the sexes. To assess if this is also the case for the *Ube3a* mouse phenotypes described above, we analyzed if there were any significant sex differences. An effect of sex was noted on the rotarod, in which female wild-type and Ube3a mice performed significantly better than male wild-type and *Ube3a* mice (p < 0.001; Fig. 2a). Since male mice are heavier than female mice and since $Ube3a^{m-/p+}$ mutants show increased weight (Fig. 2f) [8, 38], we investigated if the impaired rotarod performance as seen in $Ube3a^{m-/p}$ ⁺ mutants could be attributed to their increased weight. Hence, we performed a correlation analysis between body weight and time on the rotarod (as measured on the last training day). As shown in Fig. 2g, no meaningful correlation is observed between body weight and latency to fall in both WT mice and AS mice (WT males Pearson r = 0.08, AS males Pearson r = -0.21, WT females Pearson r = 0.35, AS females Pearson r = 0.02), although the correlation observed in WT female mice was just statistically significant (p < 0.05), indicating that increased bodyweight actually improves (rather than impairs) rotarod performance. Overall, we conclude that the impaired motor performance of $Ube3a^{m-/p+}$ mutants on the rotarod is not caused by the increased body weight observed in these mice, but truly reflects differences in motor performance.

We also observed a small effect of sex for the nest building task in which female $Ube3a^{m-/p+}$ mutants outperformed the male $Ube3a^{m-/p+}$ mutants (p < 0.05). A similar tendency was also observed in wild-type mice, but this effect was not significant (Fig. 2b). Despite the slightly better performance of female $Ube3a^{m-/p+}$ mutants, female $Ube3a^{m-/p+}$ mutants were still significantly different from wild-type mice (p < 0.001).

We observed no significant effect of sex in the open field test (p = 0.25), marble burying test (p = 0.06), and forced swim test (p = 0.27; Fig. 2c–e). Overall, these data suggest that the set of behavioral phenotypes observed in AS mice are robust and are not markedly influenced by the sex of the animal. However, given the decreased performance of male mice on the rotarod, mixed cohorts used for rotarod testing should be well balanced with respect to sex to obtain a reliable phenotype.

The behavioral test battery is suitable for within-subject testing design

A within-subject testing design is a powerful design for drug testing purposes, as it allows assessing the efficacy of a drug with considerable fewer animals. Therefore, we investigated whether the behavioral test battery allowed re-testing the same animals while maintaining a similar phenotype, which is a prerequisite for applying a within-subject design. We subjected 15 $Ube3a^{m-/p+}$ mice (Ube3a^{tm1Alb}) and 15 WT littermates in the F1 hybrid 129S2-C57BL/6 background to the behavioral test battery and repeated the test battery after a pause of 4 weeks. As shown in Fig. 3, performance on the rotarod test, nest building test, and forced swim test was highly similar when the initial test data were compared to the re-testing data. However, performance in the open field test as well as marble burying test was significantly different when this test was performed for the second time (open field: wild type initial vs retest p < 0.001, *Ube3a^m* $^{-/p+}$ initial vs retest p < 0.001; marble burying: wild type initial vs retest p < 0.001, *Ube3a*^{*m*-/*p*+} initial vs retest p < 0.0010.001; paired T test). These differences upon re-testing are likely due to the decreased anxiety levels and or habituation of the mice upon re-testing in these paradigms. Importantly, $Ube3a^{m-/p+}$ mice remained significantly different from wild-type littermates when tested for a second time, with the exception of the marble burying test, which no longer yielded a phenotype upon re-testing (p = 0.13). Hence, we conclude that most tests of the behavioral test battery are suitable for a within-subject design to test the efficacy of a drug.

Behavioral phenotypes are also observed in the *Ube3a*^{E113X} mouse model

The results above indicate that the behavioral test battery gives robust phenotypes in the $Ube3a^{tm1Alb}$ line as well as in the previously published $Ube3a^{mSTOP/p+}$ ($Ube3a^{tm1Yelg}$) line. In order to test the robustness of the battery in a third independently derived Ube3a-mutant strain, we used the $Ube3a^{mE113X/p+}$ ($Ube3a^{tm2Yelg}$) strain, which we recently described [23]. As shown in Fig. 4, the $Ube3a^{mE113X/p+}$ -mutant mice in the F1 129S2-C57BL/6J background showed again clear impairments on the rotarod test (p < 0.001), open field test (p < 0.001), marble burying test (p < 0.05), nest building test (p < 0.01), and forced swim test (p < 0.001). Taken together, these data suggest that the identified set of behavioral phenotypes in this test battery is present in three independently derived Ube3a-mutant lines.

Mouse genetic background affects the identified AS phenotypes

Previous studies have indicated the importance of the genetic background for certain *Ube3a* phenotypes [8, 9].



Fig. 2 Effect of sex on the behavioral phenotypes of *Ube3a^{m-/p+}* mice in FT hybrid 12952-C5/BL/6J background. **a** Accelerating rotarod in WT and *Ube3a^{m-/p+}* female mice (n = 52, 46) and in WT and *Ube3a^{m-/p+}* male mice (n = 68, 65). **b** Nest building test in WT and *Ube3a^{m-/p+}* female mice (n = 42, 33) and in WT and *Ube3a^{m-/p+}* male mice (n = 52, 53). **c** Open field test in WT and *Ube3a^{m-/p+}* female mice (n = 47, 41) and in WT and *Ube3a^{m-/p+}* male mice (n = 66, 65). **d** Marble burying test in WT and *Ube3a^{m-/p+}* female mice (n = 52, 46) and in WT and *Ube3a^{m-/p+}* male mice (n = 52, 46) and in WT and *Ube3a^{m-/p+}* male mice (n = 68, 65). **f** Bodyweight in WT and *Ube3a^{m-/p+}* female mice (n = 53, 50). **g** Pearson correlation test between body weight and latency to fall at day 5 in WT and *Ube3a^{m-/p+}* male mice (n = 53, 50). **g** Pearson correlation test between body weight and latency to fall at day 5 in WT and *Ube3a^{m-/p+}* female mice (n = 53, 50). **g** Pearson correlation test between body weight and latency to fall at day 5 in WT and *Ube3a^{m-/p+}* female mice (n = 53, 50). **d** Multivariate repeated ANOVA or a two-way ANOVA was used for statistical comparison. A *Bonferroni's* post hoc *test* was used to detect significant differences in behavioral phenotypes of male and female groups. All data represent mean ± SEM. Significant effects of genotype or sex are indicated as **p* < 0.05, ***p* < 0.01, and ****p* < 0.001

To test the importance of the genetic background on the behavioral test battery, we performed the test battery on AS mice on a pure C57BL/6J (Fig. 4) and 129S2 background (Fig. 5) instead of the F1 hybrid background.

 $Ube3a^{mE113X/p+}$ mice in C57BL/6J background showed a similar phenotype as $Ube3a^{mE113X/p+}$ mutants in the F1 hybrid 129S2-C57BL/6J background with respect to the rotarod test (p < 0.01), marble burying test (p < 0.001),



and nest building test (p < 0.001) (Fig. 4). No deficit was observed in the open field test (p = 0.75). Notably, the $Ube3a^{\text{mE113X/p+}}$ mice in C57BL/6J background showed a significant phenotype in the forced swim test (p < 0.05), however in the opposite direction compared to AS mice in F1 hybrid 129S2-C57BL/6J background.

The test battery was also performed using $Ube3a^{tm1Alb}$ mice in the inbred 129S2 background (Fig. 5).

 $Ube3a^{tm1Alb}$ mice in the 129S2 background did not show any of the phenotypes observed in $Ube3a^{tm1Alb}$ mice in the F1 hybrid background, with the exception of the forced swim test (p < 0.05), which yielded a similar result as obtained in mice in the F1 hybrid background. Taken together, these data confirm and extend previous studies that most AS mouse phenotypes are strongly dependent on the genetic background.



mice in F1 hybrid 129S2-C57BL/6J and C57BL/6J background. **e**–**g** Open field, marble burying, and forced swim tests in WT and *Ube3a*^{mE113X/p+} mice in F1 hybrid 129S2-C57BL/6J and C57BL/6J background. For all behavioral tests, we used a single cohort of 10 wild-type (1 female, 9 males) and 10 *Ube3a*^{mE113X/p+} mice (6 females, 4 males) in F1 hybrid 129S2-C57BL/6J, and 15 wild-type (11 females, 4 males) and 16 *Ube3a*^{mE113X/p+} (*Ube3a*^{mE113X/p+} mice (6 females, 4 males) in F1 hybrid 129S2-C57BL/6J, and 15 wild-type (11 females, 4 males) and 16 *Ube3a*^{mE113X/p+} (*Ube3a*^{mE113X/p+} (*Ube3a*^{mE113X/p+} mice (6 females, 4 males) mice in C57BL/6J background. All data represent mean ± SEM. A repeated measures ANOVA or *T* test was used for statistical comparison of genotypes, as described in the legend of Fig. 1. Significant effects of genotype are indicated as **p* < 0.05, ***p* < 0.01, and ****p* < 0.001

Susceptibility to audiogenic seizures

Epilepsy is a common feature of individuals with AS [39]. We previously showed that $Ube3a^{tm1Alb}$ mice as well as $Ube3a^{mSTOP/p+}$ ($Ube3a^{tm1Yelg}$) mice are highly susceptible to audiogenic seizures, a phenotype that is

specifically observed in mice in the 129S2 background [7]. To investigate the strength of this test in more detail, we performed a meta-analysis of five independent experiments with a total of 114 $Ube3a^{m-/p+}$ ($Ube3a^{t-m1Alb}$) mice and 45 wild-type littermates in the 129S2







background. This analysis showed that this is a very robust phenotype with seizures observed in 98% of $Ube3a^{m-/p+}$ mice and in 7% of the wild-type littermates (p < 0.001). The robustness of this test was further confirmed by a power calculation analysis (Table 2).

We tested whether seizures were also present in the $Ube3a^{\text{mE113X/p+}}$ ($Ube3a^{tm2Yelg}$) line. To that end, we crossed $Ube3a^{\text{pE113X/m+}}$ females (backcrossed eight times in 129S2) with 129S2 males. As shown in Fig. 6, an audiogenic seizure could be provoked in all $Ube3a^{\text{mE113X/p}}$ ⁺ mutants tested (p < 0.001), indicating that this phenotype is observed across three independently derived Ube3a-mutant lines.

We previously demonstrated that the sensitivity to audiogenic seizures can be reversed upon acute treatment with anti-epileptic drugs [13]. Given the high power of this assay, we investigated if this assay is suitable to determine the effective dose of a treatment. To that end, we treated mice with levetiracetam, a compound that acts as ligand of the synaptic vesicle protein 2A, which is a commonly used anti-epileptic drug for both partial and generalized seizures and which is also often prescribed to individuals with AS [40, 41]. $Ube3a^m$ $^{-/p+}$ (*Ube3a*^{tm1Alb}) mice in 129S2 background were first assessed for their (baseline) sensitivity to audiogenically evoked seizures without treatment. After establishing that all mice were sensitive, mice received at least 1 day after baseline testing a single IP dose of levetiracetam and were tested 1 h after IP injection. As shown in Fig. 6d, a good dose-response curve could be obtained, in which 2 mg/kg levetiracetam yielded approximately 60% of mice to be resistant to audiogenic seizures. This indicates that this test is highly suitable for quickly determining the effective dose of a treatment.

Minocycline treatment does not improve behavioral phenotypes of *Ube3a* mice

It has previously been reported that minocycline treatment of Ube3a animals improves synaptic plasticity as well as motor coordination, which was the basis for an open-label study with minocycline in individuals with AS (trial register NCT01531582 and [20]), as well as a randomized controlled trial ((NCT02056665), [22]). Unfortunately, the randomized trial showed no difference between placebo and minocycline-treated individuals [22]. To test if minocycline ameliorated the Ube3a-mutant phenotypes in our behavioral test battery, we subjected the animals to the same treatment protocol as used for the initial mouse study [20]. Adult-treated $Ube3a^{m-/p+}$ ($Ube3a^{t-}$ ^{m1Alb}) mice and littermate controls (8-12 weeks of age) in the F1 hybrid 129S2-C57BL/6J background received daily minocycline (45 mg/kg) or control saline IP injections starting 3 weeks prior to behavioral testing. After 3 weeks of daily injections, the mice were sequentially subjected to the behavioral test battery as described above. In contrast to the previous finding (trial register NCT01531582), we did not observe a rescue on the rotarod. We also observed no effect of minocycline on any of the other tests of the behavioral battery (Fig. 7; two-way ANOVA, genotype/treatment interaction p > 0.08 in all tests). Notably, prolonged exposure to daily minocycline injections resulted in yellow deposits over the organs and dullness of the liver (data not shown), confirming previous studies that IP administration of minocycline is not the best choice of administration [42].

Minocycline has also been used to reverse the behavioral deficits of a mouse model of Fragile X [26, 43]. Notably, in these studies, minocycline treatment was initiated immediately after birth and provided though the drinking water. Since we previously showed that a behavioral rescue of *Ube3a* mice may also depend on the timing of treatment initiation [13], we decided to treat *Ube3a* animals immediately after birth, using the same protocol as described for FMRP mice [26]. However, also this prolonged postnatal treatment regimen did not yield a significant behavioral improvement, as none of these tests showed a significant interaction of genotype and treatment (two-way ANOVA, genotype/treatment interaction p >0.16 in all tests) (Fig. 7).

Levodopa/carbidopa treatment does not improve behavioral phenotypes of *Ube3a* mice

A recent study showed that treatment of Ube3a mice with levodopa resulted in improvement of their motor skills compared to untreated Ube3a mice [21]. Based on this preclinical observation, a placebo-controlled trial of levodopa was initiated in 55 children between 4 and 12 years diagnosed with AS. Unfortunately, no significant improvement was observed on any of the outcomes measured following a 1-year treatment (trial register NCT01281475 and [21]). To test as to what extent levodopa ameliorated the phenotypes of $Ube3a^{m-/p+}$ (Ube3a^{tm1Alb}) mice in our behavioral battery, we subjected the animals to the same treatment protocol as used for the initial mouse study [21]. $Ube3a^{m-/p+}$ and wild-type littermates (8-12 weeks of age) in F1 hybrid 129S2-C57BL/6J background received daily levodopa/ carbidopa (15 mg/kg levodopa and 3.75 mg/kg carbidopa) or control saline IP injections, starting 1 h prior to behavioral testing. In contrast to the earlier finding [21], we did not observe a rescue on the rotarod. We also observed no effect of levodopa treatment on any of the other tests of the behavioral battery (two-way ANOVA, genotype/treatment interaction p > 0.17 in all tests) (Fig. 8).



Fig. 7 Effect of minocycline treatment on adult and young $Ube3a^{tm_1Alb}$ mice in F1 hybrid 129S2-C57BL/6J background. **a** Timeline representing minocycline treatment and behavioral phenotyping of adult $Ube3a^{m_-/p_+}$ mice. **b-f** Effect of minocycline on adult $Ube3a^{tm_1Alb}$ mice on the behavioral test battery. Wild-type and $Ube3a^{m_-/p_+}$ ($Ube3a^{tm_1Alb}$) vehicle-treated adult mice: n = 9, 11 (WT = 5 females, 4 males; $Ube3a^{m_-/p_+} = 6$ females, 5 males), with the exception of the nest building (n = 6, 7). Minocycline-treated wild-type and $Ube3a^{m_-/p_+}$ ($Ube3a^{tm_1Alb}$) adult mice: n = 10, 11 mice (WT = 6 females, 4 males; $Ube3a^{m_-/p_+} = 6$ females, 5 males), with the exception of the nest building (n = 6, 6). **g** Timeline representing minocycline treatment and behavioral phenotyping of young $Ube3a^{m_-/p_+}$ mice. **h**-**I** Effect of minocycline on young $Ube3a^{tm_-/p_+}$ ($Ube3a^{tm_1Alb}$) vehicle-treated young mice: n = 21, 17 (WT = 11 females, 10 males; $Ube3a^{m_-/p_+} = 7$ females, 10 males), with the exception of the nest building (n = 13, 12) and the forced swim test (20, 17). Minocycline-treated wild-type and $Ube3a^{m_-/p_+} = 4$ females, 14 males), with the exception of the open field (33, 21), the marble burying (33, 21), and the nest building (n = 16, 17). A multivariate repeated ANOVA or a two-way ANOVA was used for statistical comparison in behavioral phenotypes. *p < 0.05 and ***p < 0.001 indicate the effect of genotype. In none of the tests, we observed an interaction of genotype and treatment



Discussion

Robust behavioral phenotypes with high construct and face validity in mouse models of disease are critical for the identification of novel treatments and the successful translation of these therapies to clinical trials. These preclinical studies may give us important information about the therapeutic dose, optimal age of treatment, and the best outcome measures to be used in a clinical trial. Given the high failure rate of clinical trials aimed at improving cognitive function [44], it is absolutely critical that the preclinical data is robust (reproducible results across different mutant lines and different experimenters) and that the animal studies have high construct and face validity.

In this study, we investigated the robustness of a number of behavioral phenotypes, which we previously described using the inducible $Ube3a^{mSTOP/p+}$ ($Ube3a^{tm1Yelg}$) mice [13]. These phenotypes were assessed in two independently derived Ube3a lines: in the commonly used $Ube3a^{tm1Alb}$ line [7] and the recently generated $Ube3a^{mE113X/p+}$ ($Ube3a^{tm2Yelg}$) line [13]. Recently, we have tested two additional novel Ube3a lines in this test battery with

the same results; the $Ube3a^{tm1.1Bdph}$ line (MGI:5882092) and a novel (unpublished) Ube3a line ($Ube3a^{em1Yelg}$). Thus, taken together, a total of five independently derived Ube3a lines show phenotypes on all the behavioral tests of the test battery described in this study. In all cases, we used heterozygous Ube3a mice in which the mutation was located on the maternally inherited Ube3a allele. Therefore, we conclude that construct validity is very high. However, since the majority of individuals with AS carries a large chromosomal deletion of the AS critical region (15q11-q13) which encompasses also other genes besides Ube3a and which may contribute to a more severe phenotype [6], it would be of interest to test a mouse model of AS with large maternal deletion [11] in our behavioral test battery.

In terms of face validity, we used behavioral paradigms that assess domains of motor performance, anxiety, repetitive behavior, and seizure susceptibility, which are all relevant clinical phenotypes of AS. Nevertheless, the clinical translational value of some of our tests (e.g., open field, marble burying, nest building, and forced swim tests) may be limited. Although it is notable that many of our tests involve a strong motor component, we think that it is unlikely that the phenotypes observed in the open field, marble burying, nest building, and forced swim tests are solely related to deficits in the domain of motor functioning. Most notably, we have shown that the critical period for rescuing these phenotypes is distinctly different compared to rescuing the rotarod deficit [13] (and unpublished data). For instance, we found that gene reactivation in 3-week-old mice fully rescues the rotarod phenotype, but none of the other phenotypes [13]. It is further noticeable that both WT and mutant mice behave significantly different when tested for a second time in the open field and marble burying tests, whereas no significant changes were observed in rotarod performance. This further indicates that the deficits in the open field and marble burying tests are indicative of deficits in other domains than motor performance.

An important clinical feature of AS that is lacking in our behavioral test battery is a paradigm that assesses cognitive function. Despite profound cognitive impairments in individuals with AS, learning deficits in the AS mouse model are rather mild. We and others have reported learning deficits in AS mice by using the Morris water maze [8, 18, 45]. However, this paradigm is very labor intensive and hence less suitable for drug testing. Moreover, we found that a large number of mice are needed to detect significant differences and results varied strongly among experimenters (data not shown). A good learning paradigm that is highly suitable for drug testing is fear conditioning, in which animals are subjected to a single training session in which they are trained to associate a context (training chamber) or cue (tone) with a foot shock. However, we have not been able to get consistent results across experiments and experimenters (data not shown), and varying results are published in literature, with some studies showing a specific deficit in context conditioning [7, 46] and others a specific deficit in cued conditioning [8] or both [47-49]. Notably, the two studies that investigated the behavioral deficits of Ube3a mice across strains in great detail showed no context conditioning deficit in Ube3a mice in the F1 hybrid 129-C57BL/6J background and C57BL/ 6J background, and either normal [9] or impaired [8] cued fear conditioning in Ube3a mice in the C57BL/6J background. Collectively, these studies indicate that this phenotype is rather weak, and hence results, obtained with these tests should be interpreted with care.

By combining the data of eight independent experiments performed by five different experimenters, we were able to perform a meta-analysis of 111 $Ube3a^{m-/p+}$ (Ube3a^{tm1Alb}) and 120 WT littermate mice in the F1 hybrid 129S2-C57BL/6J background and determine the robustness of the phenotypes. In all eight experiments, we replicated *Ube3a* phenotypes observed on the rotarod test, open field test, marble burying test, nest building test, and the forced swim test. Deficits of Ube3a mice in rotarod performance, open field behavior, and marble burying have been reported by many other investigators, and hence, our results confirm the robustness of these tests. Impaired nest building behavior and impaired performance in the forced swim test of Ube3a mice have not yet been reported by other laboratories, but our study shows that these deficits are also very robust. In fact, a power analysis showed that these tests are among the most robust tests of the behavioral test battery. The open field paradigm was found to have the weakest power.

Our meta-analysis further shows that there is no major effect of sex on the behavioral phenotypes, which is in line with the general notion that such differences are also not present in AS patients. We did however find that female wild-type and mutant mice outperformed male wild-type and mutant mice on the rotarod. Improved performance of female mice on the rotarod has also been reported previously [50] and emphasizes the need of using well-matched groups when groups of both sexes of Ube3a mice are tested on the rotarod. Given that male mice are heavier than female mice, we investigated if the impaired performance of Ube3a mice on the rotarod can be attributed to the increased weight of these mutants. However, we found no correlation between weight of the animal and performance on the rotarod. This observation is in line with other studies [50-52] and indicates that the reduced performance of *Ube3a* mice on the rotarod represents a bona fide impairment in motor performance.

Besides the reproducibility of the observed phenotypes and the high face and construct validity, there are two additional features that make the behavioral test battery for *Ube3a* mice highly useful for drug testing. We show that with the exception of the epilepsy test, all behavioral experiments can be performed with a single cohort of mice, which greatly reduces costs as well as the number of mice needed. In addition, we found that with the exception of the marble burying task, the behavioral test battery can be performed twice with the same cohort while maintaining a phenotype. This makes it possible to test the efficacy of a drug using a within-subject design.

We confirmed previous studies that the audiogenic seizure phenotype is a very powerful test to investigate seizure susceptibility in Ube3a mice [7, 13, 18]. With this study, this phenotype is now also confirmed in three independently derived lines: the commonly used *Ube3a^{t-}* ^{m1Alb} line [7], the Ube3 $a^{mSTOP/p+}$ (Ube3 $a^{tm1Yelg}$) line [13], and the recently generated $Ube3a^{mE113X/p+}$ $(Ube3a^{tm2Yelg})$ line [23]. Since nearly all Ube3a mice show this phenotype compared to less than 10% of wild-type animals, this test has very high power. Moreover, we showed that the phenotype is readily reversible with the anti-epileptic drug levetiracetam and that the test is highly suitable for dose finding. The only disadvantage of the audiogenic seizure test is that it cannot be performed on the same animals as used in the behavioral test battery, since the sensitivity to audiogenic seizures is exclusively observed in Ube3a mice in the 129S2 genetic background.

We also observed an effect of genetic background on the tests of the behavioral test battery. Ube3a mice in the C57BL/6J background showed a significant phenotype in the rotarod, nest building, and marble burying tests, but no effect of genotype was observed in the open field test. A significant effect of genotype was found in the forced swim test, but remarkably, this was in the opposite direction. In contrast, Ube3a mice in the 129S2 genetic background showed only a significant deficit in the forced swim test (in the same direction as F1 hybrid mice) and no phenotype on any of the other tests of the behavioral battery. This confirms previous reports that many of the Ube3a phenotypes are very sensitive to genetic background and not present in 129 lines [8, 9]. There are however several common findings as well as a few discrepancies between these studies and our study. With respect to the rotarod [8, 9] and marble burying phenotype [9], our findings that only Ube3a-C57BL/6J and Ube3a-F1 hybrid mice show a phenotype are in full agreement with each other (Huang et al. only tested *Ube3a*-C57BL/6J in the marble burying test). With respect to the open field test (distance traveled), the other two studies also found no phenotype in Ube3a-129 mice, but in contrast to our findings, they both found a phenotype in Ube3a-C57BL/6J mice. One major difference between their and our experimental design is the time the mice were placed in the open field. Indeed, when we left the Ube3a-C57BL/6J mice for 30 min in the open field (instead of the 10 min we used), we found a nearly significant phenotype in Ube3a-C57BL/6J mice (p = 0.06; data not shown). With respect to percentage of time spent in the inner zone of the open field (which is another measure of anxiety), the other two studies showed no significant effect of genotype in any of the genetic backgrounds. Our meta-analysis did however reveal a significant difference between genotypes in F1 hybrid mice (WT 1.1% versus mutant 0.7% time in inner zone; p < 0.01), which further indicates that *Ube3a*-mutant mice are more anxious. However, we note that the observed difference was small and a significant effect was only observed in four out of the eight individual experiments. Hence, this measure is not very robust.

Taken all studies into consideration, it is clear that Ube3a mice in the F1 hybrid 129S2-C57BL/6J background show the most robust phenotypes, with the notable exception of the audiogenic seizure susceptibility test, which is strictly seen in Ube3a-129S2 mice. The question arises whether the observed differences between Ube3a mice in different genetic backgrounds have any translational significance. The lack of phenotypes of *Ube3a*-129S2 mice in most tests could simply reflect the passive/hypoactive phenotype of these mice, resulting in a floor effect. However, it could also be that the AS phenotype is sensitive to genetic background and that the changes that are observed between individuals with AS are in part caused by genetic modifiers, rather than the nature of the mutation. Detailed studies of individuals with recurrent or similar mutations could provide more insight in that question [53].

To test the translational value of the behavioral test battery, we decided to re-evaluate the two drugs that previously were tested in clinical trials involving individuals with AS: minocycline (trial register NCT01531582 [20] and NCT02056665 [22]) and levodopa (trial register NCT01281475 [21]). Both drugs were previously shown to rescue the rotarod impairment of Ube3a mice (see NCT01531582 for minocycline, and [21] for levodopa). In addition, minocycline rescued the hippocampal LTP deficit of Ube3a mice [20], whereas levodopa rescued the increased phosphorylation of CaMK2 observed in *Ube3a* mice [21]. We tested the effect of both drugs on all tests of our behavioral test battery, using the same drug administration protocols as used for the original studies. In addition, we also tested the effect of minocycline when administered from birth, as previously published

for the Fragile X mouse model [26]. However, in line with the clinical trials, we did not observe any efficacy of these drugs when tested on Ube3a mice. Our finding that minocycline and levodopa are unable to improve performance on the rotarod is at odds with aforementioned previous preclinical studies. Failure of replication could be due to differences in strains or procedures, although there is full agreement between our labs with respect to performance of Ube3a mice on the rotarod and the effects of different genetic backgrounds on this performance [9]. We think it is more likely that the rotarod experiments used for the preclinical studies were underpowered, as our analysis showed that 14 mice per group are needed for a well-powered rotarod study using two groups. In the levodopa study, the authors used 6 different treatment groups and only 6 mice per group [21]. Such small sample sizes make the test underpowered and also very vulnerable for the sex differences that we describe here. Since the details of the rotarod experiments of the minocycline treatment were not provided (NCT01531582), we cannot comment on these discrepancies.

Conclusions

Here, we provided a behavioral test battery with a robust set of well-characterized *Ube3a* phenotypes, which allows researchers to investigate the effects of pharmacological and genetic interventions involving *Ube3a* mice. A standardized set of tests, in combination with a well-defined genetic background, will also be very useful to compare data across laboratories. Moreover, using a standardized behavioral test battery may reduce selective reporting bias [54]. Future studies should reveal how well the results of this behavioral test battery can be replicated between different laboratories in which housing and testing environment is different [55–58]. In addition, robust tests that capture phenotypes in the domain of cognitive function should be identified and added to this test battery.

Abbreviations

AS: Angelman syndrome; EEG: Electroencephalography; IP: Intraperitoneal; Mut: Mutant; UBE3A: Ubiquitin-protein ligase E3A; WT: Wild-type

Acknowledgements

We thank Linda Koene for the advice concerning statistical analysis and generating the figures. We thank Maria Smit and Mireia Bernabé Kleijn for the technical assistance with behavioral experiments, Mehrnoush Aghadavoud Jolfaei for genotyping, and Minetta Elgersma-Hooisma for the mouse colony management and editing of the manuscript.

Funding

MS was supported by grants from Associazione Angelman and FROM. SSS was supported by Fundação para a Ciência e Tecnologia and Fundação Amélia de Mello. GMW was funded by the Angelman Syndrome Foundation.

Availability of data and materials

The datasets used and/or analyzed during the current study are available from the corresponding author on reasonable request.

Authors' contributions

MS, IW, SSS, JK, and DM performed the behavioral experiments. MS, IW, SSS, and YE analyzed and interpreted the data. GMW generated the *Ube3a^{tm2Yelg}* mouse model and setup the tracking system. IW made the figures. YE designed the study. MS and YE wrote the manuscript. All authors contributed intellectually to this study and edited and approved the final manuscript.

Ethics approval

All animal experiments were conducted in accordance with the European Commission Council Directive 2010/63/EU (CCD approval AVD101002016791).

Consent for publication

All authors have approved the final manuscript and consent for publication.

Competing interests

The authors declare that they have no competing interests.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Received: 1 June 2018 Accepted: 3 September 2018 Published online: 14 September 2018

References

- Petersen MB, Brøndum-Nielsen K, Hansen LK, Wulff K. Clinical, cytogenetic, and molecular diagnosis of Angelman syndrome: estimated prevalence rate in a Danish county. Am J Med Genet. 1995;60:261–2.
- Kishino T, Lalande M, Wagstaff J. UBE3A/E6-AP mutations cause Angelman syndrome. Nat Genet. 1997;15:70–3.
- Williams CA, Beaudet AL, Clayton-Smith J, Knoll JH, Kyllerman M, Laan LA, et al. Angelman syndrome 2005: updated consensus for diagnostic criteria. Am J Med Genet Part A. 2006;140A:413–8.
- Tan W-HH, Bird LM. Angelman syndrome: current and emerging therapies in 2016. Am J Med Genet Part C Semin Med Genet. 2016;401:384–401.
- Katz DM, Berger-Sweeney JE, Eubanks JH, Justice MJ, Neul JL, Pozzo-Miller L, et al. Preclinical research in Rett syndrome: setting the foundation for translational success. Dis Model Mech. 2012;5:733–45.
- Gentile JK, Tan W-H, Horowitz LT, Bacino CA, Skinner SA, Barbieri-Welge R, et al. A neurodevelopmental survey of Angelman syndrome with genotypephenotype correlations. J Dev Behav Pediatr. 2010;31:592–601.
- Jiang YH, Armstrong D, Albrecht U, Atkins CM, Noebels JL, Eichele G, et al. Mutation of the Angelman ubiquitin ligase in mice causes increased cytoplasmic p53 and deficits of contextual learning and long-term potentiation. Neuron. 1998;21:799–811.
- Huang HS, Burns AJ, Nonneman RJ, Baker LK, Riddick NV, Nikolova VD, et al. Behavioral deficits in an Angelman syndrome model: effects of genetic background and age. Behav Brain Res. 2013;243:79–90.
- Born HA, Dao AT, Levine AT, Lee WL, Mehta NM, Mehra S, et al. Straindependence of the Angelman syndrome phenotypes in Ube3a maternal deficiency mice. Sci Rep. 2017;7:1–15.
- Allensworth M, Saha A, Reiter LT, Heck DH. Normal social seeking behavior, hypoactivity and reduced exploratory range in a mouse model of Angelman syndrome. BMC Genet. 2011;12:7.
- Jiang YH, Pan Y, Zhu L, Landa L, Yoo J, Spencer C, et al. Altered ultrasonic vocalization and impaired learning and memory in Angelman syndrome mouse model with a large maternal deletion from Ube3a to Gabrb3. PLoS ONE 5(8):e12278. https://doi.org/10.1371/journal.pone.0012278.
- Miura K, Kishino T, Li E, Webber H, Dikkes P, Holmes GL, et al. Neurobehavioral and electroencephalographic abnormalities in Ube3a maternal-deficient mice. Neurobiol Dis. 2002;9:149–59.
- Silva-santos S, Van Woerden GM, Bruinsma CF, Mientjes E, Jolfaei MA, Distel B, et al. Ube3a reinstatement identifies distinct developmental windows in a murine Angelman syndrome model. J Clin Invest. 2015;125Silva-s:1–8.
- Garrido-Mesa N, Zarzuelo A, Gálvez J. Minocycline: far beyond an antibiotic. Br J Pharmacol. 2013;169:337–52.
- Elewa HF, Hilali H, Hess DC, Machado LS, Fagan SC. Minocycline for shortterm neuroprotection. Pharmacotherapy. 2006;26:515–21. https://doi.org/10. 1592/phco.26.4.515.
- 16. Harbord M. Levodopa responsive Parkinsonism in adults with Angelman syndrome. J Clin Neurosci. 2001;8:421–2.

- 17. Brown AM, Deutch AY, Colbran RJ. Dopamine depletion alters phosphorylation of striatal proteins in a model of Parkinsonism. Eur J Neurosci. 2005;22:247–56.
- Van Woerden GM, Harris KD, Hojjati MR, Gustin RM, Qiu S, Freire RDA, et al. Rescue of neurological deficits in a mouse model for Angelman syndrome by reduction of alphaCaMKII inhibitory phosphorylation. Nat Neurosci. 2007; 10:280–2.
- Weeber EJ, Jiang Y-H, Elgersma Y, Varga AW, Carrasquillo Y, Brown SE, et al. Derangements of hippocampal calcium/calmodulin-dependent protein kinase II in a mouse model for Angelman mental retardation syndrome. J Neurosci. 2003;23:2634–44. doi:23/7/2634
- Grieco JC, Ciarlone SL, Gieron-Korthals M, Schoenberg MR, Smith AG, Philpot RM, et al. An open-label pilot trial of minocycline in children as a treatment for Angelman syndrome. BMC Neurol. 2014;14:232.
- Tan WH, Bird LM, Sadhwani A, Barbieri-Welge RL, Skinner SA, Horowitz LT, et al. A randomized controlled trial of levodopa in patients with Angelman syndrome. Am J Med Genet Part A. 2018;176A:1099–107.
- Ruiz-Antorán B, Sancho López A, Cazorla-Calleja R, López-Pájaro LF, Leiva Á, Iglesias-Escalera G, et al. A randomized placebo controlled clinical trial to evaluate the efficacy and safety of minocycline in patients with Angelman syndrome (A-MANECE study). Orphanet J Rare Dis. 2018;13:144. https://doi. org/10.1186/s13023-018-0891-6.
- 23. Wang T, van Woerden GM, Elgersma Y, Borst JGG. Enhanced transmission at the calyx of Held synapse in a mouse model for Angelman syndrome. Front Cell Neurosci. 2018;11:1–19.
- Grieco J. Minocycline treatment and the necessity to develop a novel outcome measure for children with Angelman syndrome. Grad Theses Diss. 2015; http://scholarcommons.usf.edu/etd/5693.
- 25. Andes D, Craig WA. Animal model pharmacokinetics and pharmacodynamics: a critical review. Int J Antimicrob Agents. 2002;19:261–8.
- Bilousova TV, Dansie L, Ngo M, Aye J, Charles JR, Ethell DW, et al. Minocycline promotes dendritic spine maturation and improves behavioural performance in the fragile X mouse model. J Med Genet. 2009;46:94–102.
- Lee CZ, Yao JS, Huang Y, Zhai W, Liu W, Guglielmo BJ, et al. Dose–response effect of tetracyclines on cerebral matrix metalloproteinase-9 after vascular endothelial growth factor hyperstimulation. J Cereb Blood Flow Metab. 2006;26:1157–64.
- 28. Lin S, Wei X, Bales KR, Paul ABC, Ma Z, Yan G, et al. Minocycline blocks bilirubin neurotoxicity and prevents hyperbilirubinemia-induced cerebellar hypoplasia in the Gunn rat. Eur J Neurosci. 2005;22:21–7.
- Luzi P, Abraham RM, Rafi MA, Curtis M, Hooper DC, Wenger DA. Effects of treatments on inflammatory and apoptotic markers in the CNS of mice with globoid cell leukodystrophy. Brain Res. 2009;1300:146–58.
- Van Zutphen LFM, Baumans V, Beynen AC. Principles of laboratory animal science: a contribution to the humane use and care of animals and to the quality of experimental results. Amsterdam: Elsevier B.V.; 2001.
- Florek-Luszczki M, Wlaz A, Luszczki JJ. Interactions of levetiracetam with carbamazepine, phenytoin, topiramate and vigabatrin in the mouse 6 Hz psychomotor seizure model – a type II isobolographic analysis. Eur J Pharmacol. 2014;723:410–8.
- Kiel C, Faul F, Erdfelder E, Lang AG, Buchner A. G* Power 3: a flexible statistical power analysis program for the social, behavioral, and biomedical sciences. Behav Res Methods. 2007;39:175–91.
- 33. Pelc K, Cheron G, Dan B. Behavior and neuropsychiatric manifestations in Angelman syndrome. Neuropsychiatric Dis Treat. 2008;4(3):577–84.
- Kedia S, Chattarji S. Marble burying as a test of the delayed anxiogenic effects of acute immobilisation stress in mice. J Neurosci Methods. 2014;233: 150–4.
- Angoa-Pérez M, Kane MJ, Briggs DI, Francescutti DM, Kuhn DM. Marble burying and nestlet shredding as tests of repetitive, compulsive-like lehaviors in mice. J Vis Exp. 2013:1–7.
- Jirkof P. Burrowing and nest building behavior as indicators of well-being in mice. J Neurosci Methods. 2014;234:139–46. https://doi.org/10.1016/j. ineumeth.2014.02.001.
- Can A, Dao DT, Arad M, Terrillion CE, Piantadosi SC, Gould TD. The mouse forced swim test. J Vis Exp. 2011:4–8.
- Meng L, Person RE, Huang W, Zhu PJ, Costa-Mattioli M, Beaudet AL. Truncation of Ube3a-ATS unsilences paternal Ube3a and ameliorates behavioral defects in the Angelman syndrome mouse model. PLoS Genet. 2013;9(12):e1004039. https://doi.org/10.1371/journal.pgen.1004039.
- Fiumara A, Pittalà A, Cocuzza M, Sorge G. Epilepsy in patients with Angelman syndrome. Ital J Pediatr. 2010;36:31.

- 40. Weber P. Levetiracetam in nonconvulsive status epilepticus in a child with Angelman syndrome. J Child Neurol. 2010;25:393–6.
- Thibert RL, Conant KD, Braun EK, Bruno P, Said RR, Nespeca MP, et al. Epilepsy in Angelman syndrome: a questionnaire-based assessment of the natural history and current treatment options. Epilepsia. 2009;50:2369–76.
- Fagan SC, Edwards DJ, Borlongan CV, Xu L, Arora A, Feuerstein G, et al. Optimal delivery of minocycline to the brain: implication for human studies of acute neuroprotection. Exp Neurol. 2004;186:248–51.
- Rotschafer SE, Trujillo MS, Dansie LE, Ethell IM, Razak KA. Minocycline treatment reverses ultrasonic vocalization production deficit in a mouse model of Fragile X syndrome. Brain Res. 2012;1439:7–14.
- van der Vaart T, Overwater IE, Oostenbrink R, Moll HA, Elgersma Y. Treatment of cognitive deficits in genetic disorders. JAMA Neurol. 2015;72:1052.
- Daily JL, Nash K, Jinwal U, Golde T, Rogers J, Peters MM, et al. Adenoassociated virus-mediated rescue of the cognitive defects in a mouse model for Angelman syndrome. PLoS One. 2011;6:e27221.
- Hethorn WR, Ciarlone SL, Filonova I, Rogers JT, Aguirre D, Ramirez RA, et al. Reelin supplementation recovers synaptic plasticity and cognitive deficits in a mouse model for Angelman syndrome. Eur J Neurosci. 2015;41:1372–80.
- 47. Baudry M, Kramar E, Xu X, Zadran H, Moreno S, Lynch G, et al. Ampakines promote spine actin polymerization, long-term potentiation, and learning in a mouse model of Angelman syndrome. Neurobiol Dis. 2012;47:210–5.
- Sun J, Zhu G, Liu Y, Standley S, Ji A, Tunuguntla R, et al. UBE3A regulates synaptic plasticity and learning and memory by controlling SK2 channel endocytosis. Cell Rep. 2015;12:449–61. https://doi.org/10.1016/j.celrep.2015. 06.023.
- Sun J, Liu Y, Tran J, O'Neal P, Baudry M, Bi X. mTORC1–S6K1 inhibition or mTORC2 activation improves hippocampal synaptic plasticity and learning in Angelman syndrome mice. Cell Mol Life Sci. 2016;73:4303–14.
- Kovács AD, Pearce DA. Location- and sex-specific differences in weight and motor coordination in two commonly used mouse strains. Sci Rep. 2013;3:1–7.
- Cook MN, Bolivar VJ, McFadyen MP, Flaherty L. Behavioral differences among 129 substrains: implications for knockout and transgenic mice. Behav Neurosci. 2002;116:600–11.
- McFadyen MP, Kusek G, Bolivar VJ, Flaherty L. Differences among eight inbred strains of mice in motor ability and motor learning on a rotorod. Genes, Brain Behav. 2003;2:214–9.
- Abaied L, Trabelsi M, Chaabouni M, Kharrat M, Kraoua L, M'rad R, et al. A novel UBE3A truncating mutation in large Tunisian Angelman syndrome pedigree. Am J Med Genet Part A. 2010;152A:141–6.
- Tsilidis KK, Panagiotou OA, Sena ES, Aretouli E, Evangelou E, Howells DW, et al. Evaluation of excess significance bias in animal studies of neurological diseases. PLoS Biol. 2013;11:e1001609.
- 55. Mineur YS, Crusio WE. Behavioral effects of ventilated micro-environment housing in three inbred mouse strains. Physiol Behav. 2009;97:334–40.
- Richter SH, Garner JP, Würbel H. Environmental standardization: cure or cause of poor reproducibility in animal experiments? Nat Methods. 2009;6: 257–61.
- Flint J, Corley R, DeFries J, Fulker D, Gray J, Miller S, et al. A simple genetic basis for a complex psychological trait in laboratory mice. Science (80-). 1995;269:1432–5.
- Mandillo S, Tucci V, Hölter SM, Meziane H, Al BM, Kallnik M, et al. Reliability, robustness, and reproducibility in mouse behavioral phenotyping: a crosslaboratory study. Physiol Genomics. 2008;34:243–55.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

